Author’s response to reviews

Title: Prevalence and associated factors of subclinical atherosclerosis in rheumatoid arthritis at the university hospital of Kinshasa.

Authors:

Christophe Molumba (mulumbachristophe@gmail.com)
Pierrot Lebughe (lebughe7@gmail.com)
Jean-marie Mbuyi-muamba (mbuyi_muamba@yahoo.fr)
Jean-robert Makulo (jrmakulo2016@gmail.com)
François Bompeka Lepira (lepslepira@yahoo.fr)
Jean Mukaya (jmukaya@yahoo.fr)
Rene Westhovens (renewesthovens@uzleuven.be)
Patrick Verschueren (patrick.verschueren@kuleuven.be)
Jean-jacques Malemba (jeje_malemba2003@hotmail.com)

Version: 6 Date: 14 Jul 2019

Author’s response to reviews:

Detailed answers to the reviewers of manuscript BRHM-D-17-00039

Prevalence and associated factors of subclinical atherosclerosis in rheumatoid arthritis at the university hospital of Kinshasa.

We thank the editor for his comments. As suggested we have improved the following references 13,19,20,47 and 50. Moreover, we have clearly reviewed the different punctuations of all the references of this article.

Best wishes,

Dr MULUMBA